SM18 Operation Meeting (Tuesday 20 Feb  07)
Bench status  on Monday 19-Feb-2007 
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Situation on the test benches ;  

-MB1334 warmed up. 1 MB for the machine in cold test MB2394.
-All SSS tested. SSS529 : Good quench performance after the thermal cycle.
-Special tests on MB2598 (spare magnet, test with laser, MM program by Nicholas Sammut).
Situation from Cryogenics.
Situation of mechanics. 

    General Points:

· Last SM18 OP meeting.
· Test of spare magnets. All with Anti-cryostats. Standard tests + MM.
· Some spare magnets will come with no diode: No PTE test for them.
· Cryo team from 7h30 to 20 h up to the 15 March.
4  Magnet Tests completed () + 1 run on SSS529; 

2 Dipoles tests


2 SSS 500 series. 
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Quench Information
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Many Thanks To All the Teams
 For These Five Years of 

Successful Collaboration!
	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Tuesday_13_M
	TBE1
	MBBR 2424: WP07 was done in morning. The test was failed in Dipole Vs MCS Correctors. 
	ICS
	
	Trying to contact ICS repeatedly but not responding. Hence magnet waiting for stripping or standby.
Richard executed the test again: OK. Ground connection was not fixed properly.

	Tuesday_13_A
	TBE2
	MB2426 : Cyro phase lost while launching PT9 (HV after powering) and hence test could not be launched.
	ACR
	
	

	Tuesday_13_A
	TBD1
	SSS529: Load-line on MQTL. The 600 A power supply could not stand more than 100 A in negative polarity.
	MTM
	
	Guy called AB/PO. Check of the loop  OK. Ramp rate 5A/s to high for the threshold voltage of potaim card. Threshold increased and OK.

	Wednesday_14_M
	TBB1
	SSS374 : MM3 could not be launched from TEMA.
	AB/CO
	MMP
	Guy called. He reset the DIGIO card. OK.

	Wednesday_14_M

Wednesday_14_A
	TBA1
	MB1334: Problem in the Main power converter (RMA), AB/PO contacted. PT4 could not be launched.
	AB/PO
	
	Richard contacted ABPO on inferring that the circuit breakers are faulty.
RMA power converter is back in operation after that AB/PO had replaced the circuit breaker (MCB).

	Wednesday_14_A
	TBD1
	SSS529: Current automatically dropped to zero (without quench heaters being discharged, and no disturbance in cryo phase) half-way during ramp_660 (special MM).
	MTM
	
	Guy instructed to stop measurement for the day. He will check the result tomorrow morning, and will do the remaining tests.

	Monday_19_M
	TBE1
	MB2394: HV application could not be launched due to communication error with PLCSE, even after changing HV rack.
	MTM
	
	Problem rectified by resetting the PLC –HV.

	Monday_19_M
	TBE1
	MB2394: Step 'Cold temperature sensor vs ground' failed in CDBP. High leakage current detected. 
	MTM
	
	Richard detected that a pin of the temperature sensor connector inside the IFS box is bent. 
NC generated and proceeding further, as suggested by Richard.

































































